Author's response to reviews

Title: Comparison of group counseling with individual counseling in the comprehension of informed consent: a randomized controlled trial

Authors:

Rajiv Sarkar (rsarkar@cmcvellore.ac.in)
Thuppal V Sowmyanarayanan (sowmyanarayanan@cmcvellore.ac.in)
Prasanna Samuel (prasanna_samuel@hotmail.com)
Azara S Singh (azarasingh@yahoo.com)
Anuradha Bose (abose@cmcvellore.ac.in)
Jayaprakash Muliyl (jayaprakash@cmcvellore.ac.in)
Gagandeep Kang (gkang@cmcvellore.ac.in)

Version: 7 Date: 12 May 2010

Author's response to reviews:

The article cited as reference 17 has now been accepted for publication and the list of references has been changed to reflect the acceptance. A copy of the accepted paper is available.